Drivers of Digital Transformation in SMEs

Nessrine Omrani[®], Nada Rejeb[®], Adnane Maalaoui, Marina Dabić[®], and Sascha Kraus[®]

Abstract-This study aims to identify and analyze factors determining the adoption of digital technologies in SMEs. Drawing on the technology-organization-environment framework, the study highlights enabling factors from three different contexts and hypothesizes their relationship with digital technology adoption. The data used were collected from 15 346 European Union and non-European Union SMEs to test an ordered logit regression model that highlights the factors associated with an increased level of digital technologies adoption in SMEs. The empirical results show that the technology context (IT infrastructure and digital tools) along with the existing level of innovation are the main drivers that act as stepping stones in digital technology adoption. Corporate regulation, available skills, and financial resources (as organizational variables) also play a significant role in the adoption decision. Unexpectedly, the influence of the environmental context is marginal. The implications of this study are emphasized for theory and practice, laying a foundation for further empirical studies in this field.

Managerial relevance statement: This article reviews the empirical research on digital technologies adoption and examines the drivers of such adoption in SMEs. The factors identified provide guidance for practitioners adopting digital technologies in SMEs, by suggesting they assess the readiness of their firms before investing in digital technology. This research helps advance the conversation on digitalization drivers especially by bringing the discussion into the organization boundaries, as our findings highlight the predominance of organizational drivers over the technological and environmental ones. SMEs have to overcome the challenges associated with constructing an IT infrastructure capable of implementing new technologies. Indeed, while striving to adopt new digital technologies (e.g., AI, big data, IoT), many SMEs are still unprepared. Therefore, rather than adopting mimetic behaviors based on external pressure, SMEs that aim for digitalization should first assess their existing technologies, and further develop

Manuscript received 1 May 2022; revised 26 July 2022 and 20 September 2022; accepted 3 October 2022. Date of publication 18 November 2022; date of current version 16 February 2024. This work was supported in part by Slovenian Research Agency Core Project Funding "Organizing for Digital" (J5-255) and in the part by Horizon 2020 Programme of the European Union through the OpenInnoTrain project, under Grant 823971. Review of this manuscript was arranged by Department Editor D. Cetindamar. (*Corresponding author: Sascha Kraus.*)

Nessrine Omrani is with the Management and Strategy, Paris School of Business, 75013 Paris, France (e-mail: n.omrani@psbedu.paris).

Nada Rejeb is with the ICD Business School of Paris, 75010 Paris, France (e-mail: nkhachlouf@groupe-igs.fr).

Adnane Maalaoui is with the IPAG Business School, 75006 Paris, France (e-mail: a.maalaoui@ipag.fr).

Marina Dabić is with the Faculty of Economics and Business, University of Zagreb, 10000 Zagreb, Croatia, also with the University of Dubrovnik, 20000 Dubrovnik, Croatia, and also with the School of Business and Economics, University of Ljubljana, 1000 Ljubljana, Slovenia (e-mail: mdabic@efzg.hr).

Sascha Kraus is with the Faculty of Economics and Management, Free University of Bozen-Bolzano, 39100 Bolzano, Italy, and also with the Department of Business Management, University of Johannesburg, Johannesburg 2092, South Africa (e-mail: sascha.kraus@zfke.de).

Color versions of one or more figures in this article are available at https://doi.org/10.1109/TEM.2022.3215727.

Digital Object Identifier 10.1109/TEM.2022.3215727

a meticulous technological roadmap that includes skills upgrades and investments in upskilling employees' capabilities. Therefore, developing a fully integrated strategic approach is crucial before the adoption of digital technologies

Index Terms—Digital technology adoption, digital transformation, SMEs, TOE model.

I. INTRODUCTION

D IGITAL transformation has disrupted numerous markets putting immense pressure on incumbent SMEs to move toward innovation-driven organizations [70].

Digital transformation has been defined as "the use of new digital technologies, such as mobile, artificial intelligence, cloud, blockchain, and the Internet of Things (IoT) technologies, to enable major business improvements" [1, p.326]. The transition toward digital technologies offers possibilities but also challenges with its corresponding organizational changes [2] and increased systems complexity [3]. A current challenge is that digital transformation involves fundamental transformation in SMEs, which requires specific technological and organizational resources for successful adoption, e.g., knowledge management [4], information processing capability [5], and digital networking [6]. In some instances, companies need to radically change individual mindsets [7], the organizational design [8], and the corporate strategic vision while implementing modern technologies that support new business goals and customers' requirements [9].

Accordingly, academic research on the question of what drives digital technology adoption has increased in recent years. Kammerlander et al. [7] emphasized the importance of the fit between digital technologies and organization members' identity perception; for example, in terms of customer expectations. According to Li et al. [5], organizational mindfulness toward digital transformation is a prerequisite to digital technology transition. Verhoef et al. [6] argued that the strategic imperatives of digital transformation are digital resources (e.g., big data analytics capability), an organizational structure that adapts to digital change and the existence of a digital growth strategy within the company. Other studies emphasize the role of competition, as well as customers' and other stakeholders' pressure in prompting the firm's decision to adopt digital technologies [10], [11].

Despite the important advances in this research area, critical gaps remain in our understanding of digital technology adoption drivers, and it is challenging for managers to make decisions on the potential adoption of digital technologies. In particular, the academic literature on this topic is fragmented, with most studies focusing on a specific category of digital technology adoption drivers, i.e., organizational [5], technological [12], or

© 2022 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/ environmental [13]. Other studies focus only on adoption drivers of a specific technology, e.g., artificial intelligence (AI) [14], social media [15], blockchain [12], or big data [16]. While previous research has contributed to advancing knowledge in this field, focusing on a specific category of drivers and/or a specific technology would limit the generalizability of results and prevent academicians from building a theory of digital technology adoption, which is considered a milestone of digital transformation [17].

To fill these gaps, we aim to address three categories of digital technology adoption drivers: organizational, technological, and environmental, as identified by the technology–organization– environment (TOE) framework. Examining various categories of determinants would enable a better understanding of the nature of digital technology adoption drivers and evaluate the weight of each category. The core approach of this framework relies on identifying the effects of internal factors (i.e., technological and organizational) and external environmental factors. The use of an integrative model that combines various drivers and a wide range of digital technologies would contribute to building a theoretical framework on why SMEs adopt digital technologies, and whether distinctive features influence digital technology adoption in SMEs at different levels.

Data collected were used in 2020 from 15346 SMEs employing one or more persons from the EU27 and additional 12 non-EU countries. The data focus on the triggers and challenges that SMEs in Europe face when growing, transitioning to more digitalization and innovative business models. An ordered logit regression model was used to highlight the factors associated with an increased level of digital technology adoption. The findings emphasize the primacy of the technological and organizational factors. In particular, results underline the opportunities presented through digital tools exposure, firms' innovation level and the existence of corporate regulation enhancing digital technology adoption. Results also uncover the digital technology challenges induced by a shortage of both human skills and financial resources. Unexpectedly, the influence of the external environment is shown to be marginal.

This research contributes to the increasing digital transformation literature by offering an integrative view of which factors directly affect the adoption of digital technologies in SMEs. Specifically, this research provides complementary evidence on how the organizational context is combined and balanced with additional technological and (to a lesser extent) environmental contexts, to drive the digital transformation of SMEs.

As findings point to the role of internal factors (e.g., innovation rate, available skills, and internal regulation), this research points to the need to expand the theoretical basis of digital technology adoption by including theories from other fields (e.g., the resource-based view and the dynamic capabilities approach and organizational identity field), in addition to technology-based theories. In practice, most SMEs are still at an early stage of adopting advanced digital technologies as they often do not have the resources and capabilities to invest in recent technologies and need to be very efficient in allocating their financial resources [18], [19]. While the resource argument holds in the context of SMEs, other barriers should be considered by academicians and policymakers. This research has practical implications for managers in strengthening technological resources and IT infrastructure, employees' digital skills, innovation level, and corporate regulation in order to enhance firms' digital transformation.

The rest of this article is organized as follows. Section II presents the literature review and the hypotheses. Section III describes the research methodology, the data, and the model used. Section IV presents the results. Section V discusses the main findings. Finally, Section VI concludes this article.

II. LITTERATURE REVIEW

A. Theoretical Background

Technology adoption has been theorized at different levels of analysis (individual, organizational, and interorganizational). The technology acceptance model [20], the unified theory of acceptance and use of technology [21], and the theory of planned behavior [22] theorized the acceptance of information technology at the individual level. The diffusion of innovation theory focuses on both individual and organizational determinants of technology adoption. Organizational aspects include: characteristics of organizational structure, communication processes, and internal resources [23]. According to these theoretical frameworks, the technology adoption behavior is determined by the performance expectancy (perceived usefulness), effort expectancy (ease of use), and social influence. From a different perspective, the institutional theory emphasizes the role of the external environment in shaping technology adoption behavior [24]. According to the institutional theory, the adoption process is greatly affected by normative institutional pressures [13], i.e., political and social sources [25], [26], including interaction with customers, competitors, trading partners, and governments, which can potentially affect a firm's decision to adopt digital technologies.

The literature review reveals a number of studies that focused on a specific category of factors driving organization adoption behavior. Kammerlander et al. [7] examined the fit between digital technologies and organizational identity as perceived by firm members, as organizational identity might be challenged (or enhanced) by the adoption of disruptive innovation. Digital technology adoption might be contradicted, for example, by customers being conservative with regard to digitalization. According to the authors, innovation adoption is likely to occur when corresponding adaptations do not contradict their domain identity (the value delivered to customers) nor their role in the industry (leaders versus followers). Moreover, Li et al. [5] argued that organizational mindfulness (the ability of companies to capture and respond to emerging threats) raises the ability of companies to effectively deploy their organizational resources to better implement digital technologies. Verhoef et al. [6] examined the prerequisite for the digitalization of companies. They underline a number of factors: the existence of specific capabilities (digital agility and digital networking capability) and resources (digital assets), flexible and agile organizational structures that allow adaptation to change, and the use of digital platforms as a digital growth strategy. The importance of adaptation has been emphasized in innovation literature (e.g., [27]). From the same perspective, recent studies examined the organizational readiness for technology adoption defined by the degree to which an organization is ready to digitally transform the current organization [28]. Nguyen et al. [28] conceptualize readiness for technology adoption at three levels: assets, capabilities, and commitment (organizations' support for innovation initiatives). Jöhnk et al. [29] identified five AI readiness factors (strategic alignment, resources, knowledge, culture, and data). From a different perspective, previous research also shows that digital technology adoption can be driven by customer needs (improved customer service, new product offerings, tailored services, and better management of customer relationships) [2]. Some firms also use digital technologies to communicate a positive image of the firm being digital and innovative [11]. Also, suppliers as (potential) partners might be a driver for digital technology adoption. In the supply chain literature, existing research shows that firms are incited to adopt digital technologies implemented by other supply chain partners in order to optimize their operations and improve their communication [30]. Moreover, digital technologies offer the potential to help firms significantly outperform competitors by reducing their costs (e.g., reduced paperwork and error rates), optimizing their business models, improving their marketing efforts, human resource management [10], and increasing their collaboration with geographically distributed manufacturing entities [31]. As an illustration of competition pressure, Hänninen et al. [32] showed that multisided platforms are affecting the retail sector in a way that makes traditional retailers struggle to compete against this new form of business and remain relevant amidst this new digital competition.

Although this body of research has considerably improved our understanding of digital technology adoption drivers, it does not provide a full picture of the phenomena. Most works focus on either organizational, technological, or environmental factors. To overcome the subsequent fragmentation of the literature in this field, the TOE framework could be useful. The TOE model proposed by Tornatzky et al. [33] has been recognized to have superior strength over other models of technology adoption [19], [34] as it offers an integrative view of digital technology adoption drivers and allows a comparison between different categories of drivers, i.e., whether these factors affect digital technology adoption at the same level.

The TOE framework has recently been used in the context of a specific digital technology. For example, Abed [15] examined the factors that affect the intention of SMEs to use social commerce (social media and e-commerce applications) and found that the environmental context (the effect of consumer pressure and the effect of trading partners' pressure) had the most significant influence on the intention to use social commerce by SMEs, compared with organizational and technological factors. Sun et al. [35] found that the technological context (technological resources and competence) is the most influencing factor in an organization's adoption of big data. From the same perspective, El-Haddadeh et al. [16] found that technological drivers (technology perceived benefits and technology complexity) directly enhance the adoption of big data for the realization of social goals, whereas organizational and environmental factors have an indirect effect by increasing top management support for technology adoption. Based on a conceptual framework, Yang et al. [11] stated that technological intelligence (the degree of intelligence of digital technology) and supply chain cooperation are two important factors of digital technologies for supply chain management. Chen et al. [14] underlined different aspects of AI adoption that are related to AI attributes (e.g., compatibility and complexity), organizational capabilities (e.g., managerial support), as well as external environment (e.g., competitive pressure and market uncertainty). According to Liang et al. [12], the intention to adopt blockchain technology is shaped by functional and symbolic benefits, which are influenced by an existing fit between technology and environment, on the one hand, and technology and task, on the other hand. Mittal et al. [36] reviewed smart manufacturing (SM) and Industry 4.0 maturity models and investigated the triggers of SM adoption in SMEs (including the use of digital threads and big data). Research shows that the adoption of SM/Industry 4.0 depends on organizational cultural mindset change reflecting the willingness of the firm to pursue the shift toward Industry 4.0.

While combing different categories of factors to explain digital technology adoption, these studies focus on a specific technology making the generalizability of results difficult. Our study aims to fill this research gap by examining different drivers and the adoption of a wide array of digital technologies by SMEs.

In the following section, the impact of the environmental, organizational and technological contexts on the adoption of digital technologies in SMEs was hypothesized.

B. Hypotheses Development

1) Business Environment and Digital Technology Adoption: The environmental context consisting of the arena where a firm conducts its business is expected to affect the organizational decision to adopt technology innovation. The role of the external environment has been highlighted in previous research in the innovation literature [37] arguing that environmental changes, coupled with environmental uncertainty, challenge traditional business models and are critical triggers of innovation [69]. The business environment refers to external pressures and/or support for adoption, such as industry characteristics, market structure, competition [11], government regulation [13], and business infrastructure [12]. This dimension refers to the paradigmatic changes triggered by disruptive digital competitors and technologies, and changing consumer behavior [1], [38].

Collaboration with business partners is an argument cited in the literature on technology adoption. Coping with technological advances and keeping close cooperation with external partners are the main external drivers for many firms adopting digital technologies [39]. When a core player decides to use a particular digital system, other partners usually face the pressure of adapting to that system [40]. Another external driver comes from competition [6], [10]. Adopting digital technologies is regarded as an approach to enhancing firms' competitiveness. Indeed, most firms tend to adopt digital technology if they see competitors adopting it, as they believe that it is the orientation of the whole industry [10]. Many firms have been exploring how to use faster, more accurate digital management systems to facilitate lean and agile activities and stay competitive [11]. Relatedly, previous studies show that the regulatory environment at the national level can promote the development of digital technologies as entire industries, e.g., big data [35] and AI industries [14]. From this point of view, continued government support encourages firms to embrace new digital technologies.

The following is hypothesized.

Hypothesis 1: External resources available to SMEs within the business environment triggers their organizational decision to adopt digital technologies.

2) Organizational Factors and Digital Technology Adoption: The organizational context refers to the internal characteristics and resources of the firm. It is defined as formal and informal resources that reinforce the adoption of technologies, such as top management support [14], human skills and competencies [11], [41], and financial resources [29].

Digital technology adoption has strategic and operational drivers [42]. Many SMEs are actively developing digital strategies as part of their core business orientation. They believe that adopting digital technologies has the potential to trigger both incremental and disruptive innovation [11]. The adoption of digital technologies is also largely driven by internal operational problems. In particular, there is a greater need for firms to gain new business insights, unravel strategic information [2], communicate with internal and external stakeholders [43], and cut down operational costs [36].

Moreover, in order to achieve beneficial digital adoption, digital transformation requires specific organizational resources [1], [7]. Technological competencies, digital resources, support from top management, and organizational structure have been shown to influence an organization's decision to adopt digital technologies [6] and [35]. Van Zeebroeck et al. [44] observed a strong positive association between the scope of strategy change and the stage of digital technologies adoption, suggesting the existence of a link between technological structure and strategy. Besides, management support is a critical determinant of digital technology adoption. Supportive attitude and policies, allocation of financial resources [45], the existence of a strategic roadmap for digitalization processes are critical to the adoption decision. Relatedly, the adoption process usually involves changes in business tasks and processes, which might cause internal resistance. Indeed, Warner and Wäger [1] argued that rigid strategic plans, a high level of centralization and resistance to change are important internal barriers to digital transformation. Also, management support for digital security and privacy is shown to be an influential organizational factor for a firm's digital technology adoption [46].

Recently, capability building for digital transformation has received growing scholarly attention [1]. Dynamic capabilities are innovation-based and provide the capacity to create, extend, and modify a firm's resource base [47]. In particular, it has been reported that firms need to build strong dynamic capabilities to rapidly create, implement, and transform their business models to remain relevant in the emergent digital economy [48], [49]. In this perspective, digital technology adoption challenges may result from a shortage of employees and management skills [50].

In addition, regulation at the organizational level is recognized as a determinant of digital technology adoption. For example, corporate social responsibility is considered a form of corporate self-regulation that integrates into the firm's existing business models [46]. According to Ghobakhloo [46], corporate social responsibility policy ensures the firm's preservation of ethics and social norms, and clarifies the firm's relationships with its stakeholders (customers, business partners, local and global communities, etc.). Top management support plays a key role in facilitating value creation from digital technology adoption for the realization of sustainable development goals (SDGs) [16]. Some firms adopt digital technologies as part of their sustainability strategy. For example, firms utilize digital technologies (e.g., big data) to help them implement SDGs and monitor their progress [16].

Finally, an organizational context that supports innovation and fosters change is expected to enhance new technology adoption. Top and middle management can encourage innovations that further the firm's core mission and vision [51]. Their role in promoting innovation includes communication about the importance of innovation and its role within the firm's overall strategy as well as rewarding innovation behavior among subordinates.

Hypothesis 2: The availability of internal resources enhances the firm's decision to adopt digital technologies.

Hypothesis 3: The level of innovation introduced is positively associated with the adoption of digital technologies.

3) Technological Factors and Digital Technology Adoption: The technological context refers to current practices and equipment internal to the firm, as well as the set of available technologies external to the firm [24], [52]. The digital transformation offers additional relevant technologies, such as big data, blockchain, AI, machine learning, cloud computing, or social media applications.

The adoption of digital technologies depends on their perceived benefits/value. Firms adopt new information technology only if they perceive that the new technology can bring business opportunities or overcome existing deficiencies [35], [53]. According to Oliveira et al. [54], when the benefits of the technology exceed existing practices and processes, the adoption of the technology (cloud computing) is likely to occur. The perceived value of the new technology also depends on technology complexity, compatibility of the technology with the organizational previous practice [53], [55], and technology trialability, i.e., the ability to experiment with the performance of the new technology [12], [56]. At the adoption stage, perceived benefits relate to the experienced advantages and the value that have been delivered to the organization by already implemented digital technologies [57].

In addition to perceived potential advantages, the existence of an IT infrastructure and preexisting digital technology exposure are two important drivers of digital technology adoption and digital transformation [5], [16], [50]. In this perspective, cybersecurity maturity and digitalization maturity are recognized to influence the decision of SMEs to adopt digital technologies



Fig. 1. Conceptual model.

[46]. Digital maturity refers to "a firm's capability to purposefully leverage digital technologies, channels, and skills in support of transforming its processes, talent engagement, and production models" [46], p. 2392). Digital maturity enables firms to effectively benefit from digital technologies in order to reshape their organization's business models. Similarly, an integrated cybersecurity system that ensures the security, safety, and reliability of communications is indispensable and helps increase the perceived value of digital technology adoption [58]. In a similar perspective, Neumeyer et al. [59] highlighted the relevant role of technology literacy to foster technology adoption for entrepreneurship and innovation.

Overall, to achieve their full potential, digital technologies need a high level of related technology resources. SMEs with more extensive technology resources can provide a facilitating platform that smooths the adoption of new technology innovation. Technology resources refer to the existing level of IT infrastructure available to firms, such as computer hardware, software, and linkages [60]. Technology resources also include human IT resources and IT-enabled intangibles [61], which shape the speed of adoption by allowing firms to quickly reconfigure or implement the new digital technology resources [5].

Thus, the following hypothesis is proposed.

Hypothesis 4: The level of existing digital tools is positively linked to digital technology adoption.

In Fig. 1, our conceptual developments are summarized as follows. The business environment represents external triggers, including business infrastructure, availability of external resources, and access to collaboration partners, which enhance the adoption of digital technologies. The model also specifies two core categories of enablers—organizational environment (including organization internal resources, skills, corporate regulation, and innovation level) and technological resources that refer to the level of digital tools and IT infrastructure available within the firm.

III. METHODOLOGY

A. Data Collection

To test our research hypotheses, a dataset collected in Europe, namely the Eurobarometer 486, was used. The survey was conducted in the EU27 and an additional 12 non-EU countries, and focuses on the barriers and challenges that SMEs in Europe face when growing, transitioning to more digitization. The survey collected responses from 15 346 telephone interviews with SMEs employing one or more persons between February 19 and May 5, 2020.

B. Variables

1) Dependent Variable: The level of digital adoption is the dependent variable. In the Eurobarometer 486, individuals were asked, using a four-point Likert scale, about their firm's approach to digital technologies. The variable equals 3 if the firm adopts advanced technologies, 2 if it adopts basic technologies, 1 if there are difficulties to adopt digital technologies, and 0 if no digital technologies are adopted. Table I tabulates the dependent variable.

2) Independent Variables: Independent variables in our model are related to internal challenges the firm can face (financial, regulatory, innovation, access to data, payment delays, skills, etc.), the level of innovation, the business environment, and the availability of digital tools.

| TABLE I Dependent Variable | | | | | | | |
|-------------------------------|---|--|--|--|--|--|--|
| Variable | Description | | | | | | |
| Dependent variable | | | | | | | |
| Digital Adoption | Equals 3 if the enterprise adopts advanced technologies, 2 if it adopts basic technologies, 1 if there is difficulty to adopt digital technologies, and 0 if no digital technologies are adopted. | | | | | | |

Business environment: Respondents were asked to rate the business environment according to the overall strength and performance of the regional business environment, access to private and public finance, quality of support services provided by private and public actors, access to and collaboration with business partners (including other firms, public sector, educational institutions, and research organizations), availability of staff with the needed skills (including managerial skills), legal and administrative environment, and infrastructure for business, such as available office space and Internet connectivity.

Organizational environment: Respondents were asked about internal barriers to digitalization that the firms can face: the measurement items relate to innovation, regulatory obstacles or administrative burden, access to data, internationalization, access to finance, payment delays, skills including managerial skills, and difficulties with digitalization. The respondents were asked to indicate up to three key areas which represent the biggest problems for the firm.

Availability of digital tools. Respondents were asked to specify the digital technologies adopted to date: AI (e.g., machine learning or technologies identifying objects or persons), cloud computing (i.e., storing and processing files or data on remote servers hosted on the internet), robotics (i.e., robots used to automate processes, for example, in construction or design, etc.), smart devices (e.g., smart sensors, smart thermostats, etc.), big data analytics (e.g., data mining and predictive analysis), high-speed infrastructure, and blockchain.

Innovation: Respondents were asked if the firm had introduced an innovation during the past 12 months. The variable equals 1 if yes, and 0, otherwise. The innovation adopted can be: a new or significantly improved product or service to the market, a new or significantly improved production process or method, a new organization of management or a new business model, a new way of selling the goods or services, an innovation with an environmental benefit, including innovations with energy or resource efficiency benefit, and social innovation, such as new products, services or processes that aim to improve society or any other type of innovation.

3) Control Variables: The firm's size, growth, location, and country were controlled, as they have been shown to have an influence on firm innovation in previous literature (e.g., [62]). The firm's size is measured as follows: very small firm if less than ten employees (60.24% of the sample), small firm if the number of employees is between ten and 49 (21.17%), medium firm

if the number of employees is between 50 and 249 (14.41%), and big firm if the size is more than 249 (4.18%). The firm's growth is measured by the increase in the number of employees and turnover since 2016. The variable growth_employees takes the value 0 if the number of employees has decreased, 1 if the number of employees remains stable, 2 if it has grown by less than 30%, and 3 if it has grown by at least 30%. The variable growth_turnover takes the value 0 if the turnover has decreased, 1 if the turnover remains stable, 2 if it has grown by less than 30%, and 3 if it has grown by at least 30%. A set of mutually exclusive binary variables are used for a geographic area: large town/small town and rural area/industrial area.

Tables II and III provide, respectively, a summary of the independent variables used in the model and the descriptive statistics of dependent and independent variables.

C. Model

The data analysis has followed different stages. A descriptive analysis and a univariate analysis were performed to explore the sample profile, and then multivariate analysis techniques and ordered logit regression were applied to address the research questions.

The steps involved in data analyses using Stata Software are introduced in the following.

1) Ordered Logit Model: Because the dependent variable is ordered (no adoption, difficulty to adopt, adoption of basic technologies, and adoption of advanced technologies), an ordered logit regression was used to estimate the following models:

$$Y_i = \text{Business}_i \ \beta_1 + \ \varepsilon_i \tag{1}$$

where Y_i is the level of digital adoption of firm *i*, Business_{*i*} indicates the rating of the business environment, β is the vector of coefficients, and ε_i is a normally distributed random error term.

$$Y_i = \text{Internal}_i \ \beta_1 + \ \varepsilon_i \tag{2}$$

where Y_i is the level of digital adoption of firm *i*, Internal_{*i*} indicates the areas that pose problems for the enterprise, β is the vector of coefficients, and ε_i is a normally distributed random error term.

$$Y_i = \text{Innovation}_i \ \beta_1 + \ \varepsilon_i \tag{3}$$

TABLE II DESCRIPTIONS OF INDEPENDENT VARIABLES

| Variable | Description | | | | | | | | |
|----------------------|--|--|--|--|--|--|--|--|--|
| 1 | | | | | | | | | |
| Business environment | Rating the business environment according to eight measures. | | | | | | | | |
| Internal environment | Eight key areas that pose the biggest problems for the enterprise. | | | | | | | | |
| Innovation | The introduction of an innovation during the past 12 months according to six types of innovation. | | | | | | | | |
| Digital tools | Type of digital technologies adopted. | | | | | | | | |
| Control variables | | | | | | | | | |
| Size | The size of the firm, three dummy variables. | | | | | | | | |
| Location | Four dummy variables: large town or a small town, in a rural or an industrial area. | | | | | | | | |
| | | | | | | | | | |
| Growth | The growth of the number of employees and turnover since 2016. The variable growth_employees takes the values from 0 to 2. | | | | | | | | |
| Country | In total, 39 dummy variables referring to the 27 European Union countries and additional 12 non-EU countries and territories | | | | | | | | |

where Y_i is the level of digital adoption of firm *i*, Innovation_{*i*} indicates the type of innovation introduced during the last year, β is the vector of coefficients, and ε_i is a normally distributed random error term.

$$Y_i = \text{Tools}_i \beta_1 + \varepsilon_i \tag{4}$$

where Y_i is the level of digital adoption of firm *i*, Tools_{*i*} indicates the type of digital technologies adoption, β is the vector of coefficients, and ε_i is a normally distributed random error term.

$$Y_{i} = \text{Business}_{i} \ \beta_{1} + \text{Internal}_{i}\beta_{2} + \text{Innovation}_{i}\beta_{3}$$
$$+ \text{Tools}_{i}\beta_{4} + \varepsilon_{i}$$
(5)

where Y_i is the level of digital adoption of firm *i*, Business_{*i*} indicates the rating of the business environment, $Internal_i$ indicates the areas that pose problems for the enterprise, Innovation_i indicates the type of innovation introduced during the last year, Tools_i indicates the type of digital technologies adoption, β is the vector of coefficients, and ε_i is a normally distributed random error term.

$$Y_{i} = \text{Business}_{i} \beta_{1} + \text{Internal}_{i}\beta_{2} + \text{Innovation}_{i}\beta_{3} + \text{Tools}_{i}\beta_{4} + X_{i}\beta_{5} + \varepsilon_{i}$$
(6)

where Y_i is the level of digital adoption of firm *i*, Business_{*i*} indicates the rating of the business environment, $Internal_i$ indicates the areas that pose problems for the enterprise, Innovation $_i$ indicates the type of innovation introduced during the last year, Tools_{*i*} indicates the type of digital technologies adoption, X_i is a vector containing the values of the size, the location, and the growth of the firm, β is the vector of coefficients, and ε_i is a normally distributed random error term.

.

$$Y_{i} = \text{Business}_{i} \beta_{1} + \text{Internal}_{i}\beta_{2} + \text{Innovation}_{i}\beta_{3} + \text{Tools}_{i}\beta_{4} + Z_{i}\beta_{5} + \varepsilon_{i}$$
(7)

where Y_i is the level of digital adoption of firm *i*, Business_{*i*} indicates the rating of the business environment, $Internal_i$ indicates the areas that pose problems for the enterprise, Innovation $_i$ indicates the type of innovation introduced during the last year, Tools_i indicates the type of digital technologies adoption, Z_i is a vector containing the values of the size, the location, the growth, and the country of the firm, β is the vector of coefficients, and ε_i is a normally distributed random error term.

IV. RESULTS

The results of the logistic regressions are reported in Table IV. Because cross-sectional data were relied upon, it was not possible to account for endogeneity among the variables and the links between variables are correlational. The relationship between variables is explained instead of causality. To ensure the robustness of our results, seven model specifications were performed, showing the results in blocks. The largest VIF ranged from 1.01 to 2.02, and the average VIF ranged from 1.01 to 1.56. Since none of the VIF values exceeds 10 and none of the average VIF exceeds 6, it was concluded that there was no multi-collinearity problem.

| Type of variable | Description | Mean | Std dev. | Min. | Max. |
|-----------------------|---|--|--|--|--|
| Dependent variables | | | | | |
| Digital Adoption | | 1.732 | 1.053 | 0 | 3 |
| Independent variables | | | | | |
| Business environment | All Finance Support Partners Skills Sustainability Legal Infrastructure | 2.004 1.781 1.767 1.976 1.684 1.599 1.873 2.216 | .693 .796 .745 .694 .859 .811 .757 .695 | 0 0 0 0 0 0 0 0 0 | 3 3 3 3 3 3 3 3 3 3 |
| Internal environment | Innovation issue Regulation Data Access Internationalization Finance Payment issue Skills Digitalization issue | .099 .467 .068 .078 .188 .325 .214 .111 | .299 .498 .252 .268 .390 .468 .410 .315 | 0 0 0 0 0 0 0 0 0 0 | 1 1 1 1 1 1 1 1 1 |
| Innovation | | .615 | .486 | 0 | 1 |
| Digital tools | AI Cloud Robotics Smart Big Data Infrastructure Blockchain | .076 .478 .085 .277 .144 .337 .033 | .265 .499 .279 .448 .351 .472 .178 | 0 0 0 0 0 0 0 | 1 1 1 1 1 1 1 |
| Size | | .625 | .879 | 0 | 3 |
| Location | Large town Small town Rural Industrial | .491 .356 .102 .128 | .499 .479 .303 .334 | 0 0 0 0 | 1 1 1 1 |
| Growth | Employees Turnover | 1.567 1.581 | .855 .983 | 0 0 | 3 3 |

TABLE III DESCRIPTIVE STATISTICS

The first model gives results using the business environment as the only independent variable. The second model introduces internal environment variables measures. In the third model, the link of digital adoption with the innovation variable was measured. In the fourth model, digital tools available are included. In the fifth model, all the variables of interest are tested. In the sixth model, the control variables are added, and in the last model, the country is added. Empirical findings from these models show substantial robustness across the seven performed model specifications.

According to Table IV, Model (1) shows that regarding the business environment, finance ($\beta = .076, z = 3.59; p < .01$), partners ($\beta = .049, z = 2.10; p < .05$), skills ($\beta = -.047, z = -2.55; p < .05$), and infrastructure ($\beta = .122, z = 5.47; p < .01$)

TABLE IV DIGITAL ADOPTION—ORDERED LOGISTIC REGRESSION RESULTS

| | (1) | | (2) | | (3) (4) | | (5) | | | (6) | | (7) | | |
|---|---|--|-------------------------------|-------------------------------|-------------------|---------|---|---|---|---|---|---|---|---|
| Business | Digital Ad | option | Digital Ad | option | Digital Ac | loption | Digital A | doption | Digital A | aoption | Digital | Adoption | Digital 2 | Adoption |
| Environment All Finance Support Partners Skills Sustainable Legal Infrastructure Internal | $\begin{array}{c} 0.039 \\ 0.077^{***} \\ 0.020 \\ 0.049^{**} \\ -0.048^{**} \\ 0.050^{**} \\ 0.005 \\ 0.123^{***} \end{array}$ | (0.024) (0.021) (0.024) (0.023) (0.019) (0.022) (0.023) (0.022) | | | | | | | -0.029 0.006 0.021 0.025 -0.029 0.018 0.006 0.068*** | $\begin{array}{c} (0.025) \\ (0.022) \\ (0.025) \\ (0.024) \\ (0.020) \\ (0.023) \\ (0.024) \\ (0.023) \end{array}$ | -0.042* -0.002 0.019 0.021 -0.032 0.015 0.006 0.064*** | $\begin{array}{c} (0.025) \\ (0.023) \\ (0.025) \\ (0.024) \\ (0.020) \\ (0.023) \\ (0.024) \\ (0.023) \end{array}$ | -0.034 0.016 0.027 0.002 -0.000 0.022 -0.009 0.044* | (0.026) (0.023) (0.025) (0.025) (0.021) (0.023) (0.025) (0.024) |
| Environment Innovation issue | | | 0.027 | (0.048) | | | | | -0.066 | (0.048) | -0.074 | (0.048) | -0.003 | (0.049) |
| Regulation Data access Internationali | | | 0.168*** 0.078 0.219*** | (0.029) (0.061) (0.057) | | | | | 0.084*** -0.102 0.044 | (0.031) (0.063) (0.057) | 0.076** -0.104* 0.040 | (0.031) (0.063) (0.058) | 0.086*** -0.078 0.078 | (0.032) (0.064) (0.058) |
| zation. Finance Payment | | | -0.215*** -0.045 | (0.036) (0.031) | | | | | -0.189*** -0.029 | (0.039) (0.032) | -0.182*** -0.023 | (0.039) (0.032) | -0.095** -0.017 | (0.040) (0.032) |
| Skills Digitalization issue | | | 0.209*** -0.002 | (0.037) (0.045) | | | | | 0.067* 0.007 | (0.039) (0.046) | 0.049 -0.001 | (0.039) (0.047) | 0.084** 0.009 | (0.041) (0.047) |
| Innovation Digital tools | | | | | 0.576*** | (0.029) | | | 0.296*** | (0.031) | 0.282*** | (0.031) | 0.302*** | (0.032) |
| AI Cloud Robotics Smart Big Data Infrastructure Blockchain Size | | | | | | | 0.489*** 0.511*** 0.454*** 0.315*** 0.498*** 0.400*** 0.235** | (0.071) (0.032) (0.065) (0.037) (0.052) (0.035) (0.098) | 0.463*** 0.468*** 0.399*** 0.271*** 0.461*** 0.372*** 0.225** | $\begin{array}{c} (0.071) \\ (0.032) \\ (0.066) \\ (0.037) \\ (0.052) \\ (0.035) \\ (0.097) \end{array}$ | 0.457*** 0.453*** 0.335*** 0.420*** 0.420*** 0.368*** 0.223** 0.126*** | $\begin{array}{c} (0.071) \\ (0.032) \\ (0.066) \\ (0.038) \\ (0.053) \\ (0.035) \\ (0.098) \\ (0.019) \end{array}$ | 0.466*** 0.441*** 0.320*** 0.267*** 0.468*** 0.390*** 0.240** 0.129*** | (0.072) (0.033) (0.067) (0.038) (0.054) (0.037) (0.099) (0.019) |
| Growth Employees Turnover | | | | | | | | | | | 0.003 0.043** | (0.018) (0.017) | 0.001 0.036** | (0.019) (0.017) |
| Small town Large town Rural Industrial | | | | | | | | | | | Ref. 0.029 Ref. 0.011 | (0.030) | Ref. 0.088*** Ref. 0.028 | (0.032) |
| Country France Belgium Netherlands Germany Italy Luxembourg Denmark Ireland UK Greece Spain Portugal Finland Sweden Austria Cyprus Czech | | | | | | | | | | | | | $\begin{array}{c} Ref: \\ -0.107 \\ 0.260 \ast \\ 0.147 \\ 0.488 \ast \ast \\ -0.339 \ast \\ -0.201 \\ -0.273 \ast \\ -0.201 \\ 0.373 \ast \\ -0.347 \ast \ast \\ 0.309 \ast \ast \\ 0.169 \\ -0.229 \\ -0.491 \ast \ast \end{array}$ | (0.126) (0.125) (0.132) (0.114) (0.184) (0.134) (0.137) (0.134) (0.135) (0.129) (0.131) (0.125) (0.161) (0.138) |
| Kepublic Estonia Hungary Latvia Lithuania Malta Poland Slovakia Slovakia Slovania Slovania Slovania Slovania Romania Turkey Croatia Makedonia Serbia Norway Iceland Japan USA Brazil Bosnia and Herzegovina | | | | | | | | | | | | | 0.300*** 0.143 -0.343*** -0.139 0.254 0.069 -0.131 0.215* -0.202* -0.125 0.207* -0.113 0.327*** 0.507*** 0.507*** 0.507*** 0.408*** -0.408*** | (0.109) (0.132) (0.124) (0.127) (0.164) (0.125) (0.128) (0.129) (0.121) (0.123) (0.131) (0.147) (0.122) (0.166) (0.131) (0.144) (0.132) |
| Canada cut1 | | | | | | | | | | | | | -0.468*** | (0.125) (0.137) |
| cons cut2 | -0.905*** | (0.066) | -1.435*** | (0.033) | -1.184*** | (0.028) | -1.001*** | (0.025) | -0.702*** | (0.079) | -0.651*** | (0.082) | -0.618*** | (0.123) |
| cons cut3 | 0.070 | (0.066) | -0.462*** | (0.028) | -0.206*** | (0.024) | 0.000 | (0.022) | 0.304*** | (0.077) | 0.356*** | (0.081) | 0.412*** | (0.122) |
| cons | 1.589*** 15346 | (0.068) | 1.055*** 15346 | (0.028) | 1.323*** 15346 | (0.025) | 1.624*** 15346 | (0.025) | 1.939*** 15346 | (0.078) | 1.994*** 15346 | (0.082) | 2.079*** 15346 | (0.123) |

Standard errors in parentheses * p<.10, ** p<.05, *** p<.01

have significant link with digital adoption meaning that business environment is linked to digital adoption. Model (2) reveals significant links of regulation ($\beta = .168, z = 5.73; p < .01$), internationalization ($\beta = .219, z = 3.82; p < .01$), finance ($\beta = -.214, z = -5.93; p < .01$), and skills ($\beta = .208, z = 5.71; p < .01$) with digital adoption reflecting that some internal factors are linked to digital adoption. Model (3) indicates a significant link between innovation and digital adoption. Model (4) reveals a significant link between all the digital tools and digital adoption.

Model (7) shows that, concerning the impact of the business environment context, only business infrastructure has a positive and significant predictive power of digital adoption ($\beta = .043$, z = 1.82; p < .1; meaning that the business infrastructure is the most relevant external trigger of the adoption decision. As far as the internal environment is concerned, the regression model results show that corporate regulation ($\beta = .086$, z = 2.66; p < .01) and human skills ($\beta = .084, z = 2.08; p < .05$) have positive and significant links with digital adoption, emphasizing that firms adopting the most advanced digital technologies have a high level of skills to deal with new technologies, as well as internal laws/norms that enhance new digital technology adoption. Financial difficulties are negatively linked to the dependent variable adoption ($\beta = -.095$, z = -2.40; p < .05) meaning that the lack of financial resources is one of the most important barriers to the adoption decision. The innovation level is positively linked to digital technology adoption ($\beta = .302, z = 9.45; p < .01$), showing that firms adopting digital technologies are those who have the highest rates of innovation. Finally, digital tools are positively linked to the dependent variable adoption: AI ($\beta = .466$, z = 6.47; p < .01), cloud ($\beta = .440, z = 13.27; p < .01$), robotics $(\beta = .319, z = 4.76; p < .01)$, smart $(\beta = .266, z = 6.94; p < .01)$, big data (β = .467, z = 8.61; p < .01), infrastructure (β = .389, z = 10.40; p < .01), and blockchain ($\beta = .240, z = 2.43; p < .05$). This result shows that firms who are familiar with digital tools are more likely to adopt recent digital technologies. In particular, technology infrastructure, as a technology resource, has been shown in our findings to be an important driver of digital technology adoption. Digital technology infrastructure aims to secure the firm's information, ensure higher data processing capacity, and enable a superior overall technology that is appropriate for the firm's business [63]. At a strategic level, digital technology infrastructure includes the existence of policies that secure the integration of digital technologies throughout the organization [46].

Taken together, the results of this study demonstrate that digital technology adoption is predominantly shaped by two contexts: the technological and the organizational. The environmental context is shown to have little influence on the adoption decision.

With regard to the firm size, results show a positive and significant link. This result corroborates studies arguing that larger firms are more likely to adopt technology innovation because they have more resources and capabilities to support the digital technologies adoption process [64]. Larger companies can absorb risks and initial costs [65], [66], and tend to be innovators in technology [67]. Results also indicate that digital technologies have more potential to be adopted by firms who are located in a large town (as compared with those in a small town), and by those who have the greatest turnover. Finally, it was found that, in comparison to the reference country - France, firms operating in The Netherlands, Italy, Finland, Sweden, Estonia, Croatia, Serbia, Norway, and Bosnia and Herzegovina adopt more advanced digital technologies, whereas individuals living in Denmark, Ireland, Greece, Spain, Czech Republic, Latvia, Bulgaria, Romania, Japan, USA, Brazil, and Canada adopt less advanced digital technologies.

V. DISCUSSION

Decisions regarding whether or not to adopt digital technologies are of strategic importance to SMEs' growth and survival. Our paper aims to offer an integrative view of what factors directly affect the adoption of digital technologies in SMEs while considering a wide range of technologies.

Our findings show that digital technology adoption is prominently driven by internal factors. Indeed, the adoption decision is dependent foremost on the technological context (the existing IT infrastructure and previous digital technology exposure of organization members), followed by the organizational context (i.e., innovation rate, employee skills, corporate regulation, and financial resources). The external environment is shown to have the lowest impact on the level of digital technology adoption. Indeed, only business infrastructure is perceived as a significant driver. No significant evidence was found of the impact of pressure from external partners. This finding is counterintuitive, as previous research underlined the relevance of external partners (e.g., customers and competitors) in the decision to adopt specific digital technologies, e.g., blockchain [12] and social media [15]. One explanation for this divergent result could be that the adoption of certain types of technologies may occur under external pressure, given, for example, data security concerns (e.g., blockchain) or the emergence of disruptive competing business models (e.g., social media commerce).

Furthermore, among organizational drivers, the significant impact of corporate regulation (internal norms and laws) is particularly relevant as this finding is in line with increasing research emphasizing the importance of cultural and identity aspects when adopting digital technologies, e.g., culture of innovativeness [29], role and domain identity (self-image) that makes the organization different from the others in the eyes of its members [7], and a culture of knowledge sharing [36]. Corporate regulation has been associated with corporate social responsibility to the adoption of Industry 4.0 technologies [46], which offer immense opportunities for the realization of sustainable manufacturing. Our findings echo this argument and lend empirical support to corporate regulation as a driver of digital technology adoption. Also, the positive association between employee skills and financial resources, on the one hand, and technology adoption, on the other hand, is supported by previous work highlighting the importance of digital assets [6] and capabilities [1], [14] for firms aiming to transform digitally.

Interestingly, the most significant association among organizational drivers is between the existing level of innovation and digital technology adoption. At a similarly high level is the association between the technological drivers (i.e., existing level of digital tools and IT infrastructure) and digital technology adoption. This result offers relevant insights into the investigated topic. Indeed, this result suggests that the main driver of technology adoption is "organization preparedness" for digitalization. This result aligns with previous studies highlighting the prominence of the technological and organizational aspects in "organization readiness" to digital technology adoption, defined by the degree to which an organization is ready to digitally transform the current organization (e.g., [29], [50], and [68]). This body of research put forward the relevance of available resources and capabilities, knowledge, culture, existing data, and the strategic alignment between technology adoption and the business strategy.

In light of the discussed literature and our findings previously described, it might be the case that companies in specific contexts

may be prompted to adopt digital technologies under external stakeholders' pressure in order to be competitive [12], [15]. This may appear irrational and driven by mimetic behavior. Therefore, previous studies, as well as the results of the current research, refer to the importance of digital technology readiness that depends on technological and organizational factors (see [29] and [50]). As the digital adoption of companies in our sample is driven by organizational and technological aspects, a high degree of digital would be expected readiness for those companies. In order to further advance our understanding of digital technology adoption, it would be interesting to compare the success of the adoption decision in firms having different drivers (for instance external versus internal). This would inform managers about the relevance and the timeliness of their adoption decision. Longitudinal studies would be helpful for this purpose.

A. Implications for Theory

This study makes three contributions to digital transformation literature. First, it contributes to advancing the conversation related to factors that drive the organization's decision to adopt digital technologies. Although previous research has considerably improved our knowledge on this topic, most studies only provide a fragmented picture of the phenomena by focusing either on organizational, technological, or environmental factors. By considering different categories of drivers, it was possible to notice the important weight of internal factors compared with external ones. It was possible to demonstrate that the adoption decision is foremost dependent on the resources and capabilities available within the firm. These findings bring the discussion on digitalization drivers into organization boundaries and open the way for cross-fertilization between technology adoption theories and other organizational theories (e.g., the resourcebased view and the dynamic capabilities approach) in order to extend our understanding of the digital transformation process. The need for more theories in the digital technology field is recently underlined by academicians. Indeed, a recent systematic literature review on digital technology adoption reveals that amongst the selected 55 papers, only 22 of them discussed their studies from a theoretical perspective [11]. This gap indicates that more theoretical underpinnings are required in the field. Second, contrary to previous research which has focused on a unique digital technology (e.g., [12] and [35]), a large range of technologies that are increasingly implemented in SMEs (AI, blockchain, cloud, etc.) was targeted. This would allow the generalizability of the results and ultimately the building of a theory of digital technology adoption. Third, quantitative analysis from a large dataset across different EU and non-EU countries was used. Our quantitative large-scale panel dataset allowed us to complement existing qualitative and conceptual work, which is likely to provide a better understanding of the digital technology adoption decision.

B. Practical Implications

Overall, the factors identified provide guidance for practitioners adopting digital technologies in SMEs, by suggesting they assess the readiness of their firms before investing in digital technology. For example, our findings indicate that the potential benefit of a preexisting IT infrastructure for new digital technology adoption has been realized by the participants in this study. SMEs have to overcome the challenges associated with constructing an IT infrastructure capable of implementing new technologies. Indeed, while striving to adopt new digital technologies (e.g., AI, Big data, IoT), many SMEs are still unprepared. Therefore, rather than adopting mimetic behaviors based on external pressure, SMEs that aim for digitalization should first assess their existing technologies, and further develop a meticulous technological roadmap that includes skills upgrades and investments in upskilling employees' capabilities. Therefore, developing a fully integrated strategic approach is crucial before the adoption of digital technologies. Finally, our results highlight the need for firms to help their people embrace the right behaviors and values for working in a digital environment.

In response to these challenges, national and international associations must develop digital technology supportive policies, similar to what the OECD has initiated via the OECD Going Digital Project awareness-raising program¹ to stimulate the digitalization of SMEs. Although the basic level of technology resources, especially IT structure, could be sufficient for many SMEs seeking to adopt recent digital technologies, our findings suggest that firms should continue to build their technological resources and competencies. For this purpose, more employees with digital-related knowledge need to be recruited to facilitate the adoption and implementation of new technologies. Firms should also rise the digital awareness of their employees by communicating values and regulations that enhance the adoption and use of new technologies. Finally, SMEs should be able to find the needed financial resources and prioritize their digital transformation strategies, despite their resource constraints.

VI. CONCLUSION

Our current understanding of what drives SMEs' digital transformation is limited by the lack of integrative research that covers issues beyond contextual imperatives (i.e., economic, technological, legal, and financial infrastructure). Using a quantitative large-scale panel dataset across different EU and non-EU countries, this study intended to complement the existing qualitative and conceptual works on digital technology adoption drivers. Investigating a variety of categories of drivers allowed us to overcome the fragmented picture provided by previous research that focused on separate aspects of the adoption decision. Our research helps advance the conversation on digitalization drivers especially by bringing the discussion into the organization boundaries, as our findings highlight the predominance of organizational drivers over the technological and environmental ones. The engaged discussion opens the way for cross-fertilization between technology adoption theories and other organizational theories (e.g., the resource-based view and the dynamic capabilities approach) in order to extend our understanding of the digital transformation process.

¹[Online]. Available: https://www.oecd.org/digital/going-digital-project/

As with every research, of course also this study is associated with several limitations opening opportunities for further research that replicates and/or extends this study. Due to the cross-sectional nature of data, it was not possible to identify the causal relationship between environmental, organizational, technological contexts, and digital technology adoption. Future research should further study causality and identify heterogeneous digital adoption behavior between SMEs. Also, further longitudinal research is needed that examines the influence of drivers of digital transformation beyond the adoption decision (e.g., by assessing the adoption success). In addition, longitudinal studies would help understand the evolution of digital technology drivers over time. Companies may be prompted to adopt a digital technology under external pressure; however, the adoption success will be contingent on their organizational and technological readiness. In this perspective, "organizational readiness" could be introduced as a moderator factor in order to offer further insights on digital technology adoption outcomes.

Moreover, the identified digital technology adoption drivers in this study provide a theoretical basis for further investigating the intention to adopt digital technologies within SMEs. It is deemed that future research can use the resource-based view and dynamic capabilities theories to investigate skills and resources to improve the capabilities of successful adoption of digital technologies. Finally, the results of this study highlight the need for further research on how corporate social responsibility influences the organization's decision on digital technology adoption.

ACKNOWLEDGMENT

The content of this publication does not reflect the official opinion of the European Union. Responsibility for the information and views expressed in this publication lies entirely with the authors.

REFERENCES

- K. S. Warner and M. Wäger, "Building dynamic capabilities for digital transformation: An ongoing process of strategic renewal," *Long Range Plan.*, vol. 52, no. 3, pp. 326–349, 2019.
- [2] S. Kraus, P. Jones, N. Kailer, A. Weinmann, N. Chaparro-Banegas, and N. Roig-Tierno, "Digital transformation: An overview of the current state of the art of research," SAGE Open, vol. 11, no. 3, pp. 1–15, 2021.
- [3] N. Lakemond, G. Holmberg, and A. Pettersson, "Digital transformation in complex systems," *IEEE Trans. Eng. Manage.*, early access, Oct. 26, 2021, doi: 10.5465/AMBPP.2020.19455abstract.
- [4] M. F. Manesh, M. M. Pellegrini, G. Marzi, and M. Dabic, "Knowledge management in the fourth industrial revolution: Mapping the literature and scoping future avenues," *IEEE Trans. Eng. Manage.*, vol. 68, no. 1, pp. 289–300, Feb. 2021.
- [5] H. Li, Wu Y., D. Cao, and Y. Wang, "Organizational mindfulness towards digital transformation as a prerequisite of information processing capability to achieve market agility," *J. Bus. Res.*, vol. no. 122, pp. 700–712, 2021.
- [6] P. C. Verhoef et al., "Digital transformation: A multidisciplinary reflection and research agenda," J. Bus. Res., vol. 122, pp. 889–901, 2021.
- [7] N. Kammerlander, A. König, and M. Richards, "Why do incumbents respond heterogeneously to disruptive innovations? The interplay of domain identity and role identity," *J. Manage. Stud.*, vol. 55, no. 7, pp. 1122–1165, 2018.
- [8] T. Kretschmer and P. Khashabi, "Digital transformation and organization design: An integrated approach," *California Manage. Rev.*, vol. 62, no. 4, pp. 86–104, 2020.

- [9] T. Niemand, J. P. Coen Rigtering, A. Kallmünzer, S. Kraus, and A. Maalaoui, "Digitalization in the financial industry: A contingency approach of entrepreneurial orientation and strategic vision on digitalization," *Eur. Manage. J.*, vol. 39, no. 3, pp. 317–326, 2021.
- [10] D. Q. Chen, D. S. Preston, and M. Swink, "How the use of big data analytics affects value creation in supply chain management," *J. Manage. Inf. Syst.*, vol. 32, no. 4, pp. 4–39, 2015.
- [11] M. Yang, M. Fu, and Z. Zhang, "The adoption of digital technologies in supply chains: Drivers, process and impact," *Technological Forecasting Social Change*, vol. 169, 2021, Art. no. 120795.
- [12] T. P. Liang, R. Kohli, H. C. Huang, and Z. L. Li, "What drives the adoption of the blockchain technology? A fit-viability perspective," *J. Manage. Inf. Syst.*, vol. 38, no. 2, pp. 314–337, 2021.
- [13] S. A. Sherer, C. D. Meyerhoefer, and L. Peng, "Applying institutional theory to the adoption of electronic health records in the US," *Inf. Manage.*, vol. 53, no. 5, pp. 570–580, 2016.
- [14] H. Chen, L. Li, and Y. Chen, "Explore success factors that impact artificial intelligence adoption on telecom industry in China," *J. Manage. Analytics*, vol. 8, no. 1, pp. 36–68, 2021.
- [15] S. S. Abed, "Social commerce adoption using TOE framework: An empirical investigation of Saudi Arabian SMEs," *Int. J. Inf. Manage.*, vol. 53, 2020, Art. no. 102118.
- [16] R. El-Haddadeh, M. Osmani, N. Hindi, and A. Fadlalla, "Value creation for realizing the sustainable development goals: Fostering organizational adoption of big data analytics," *J. Bus. Res.*, vol. 131, pp. 402–410, 2021.
- [17] J. Bughin, T. Kretschmer, and N. van Zeebroeck, "Digital technology adoption drives strategic renewal for successful digital transformation," *IEEE Eng. Manage. Rev.*, vol. 49, no. 3, pp. 103–108, Sep. 2021.
- [18] K. Bär, Z. N. L. Herbert-Hansen, and W. Khalid, "Considering industry 4.0 aspects in the supply chain for an SME," *Prod. Eng.*, vol. 12, no. 6, pp. 747–758, 2018.
- [19] S. A. Qalati, W. Li, E. G. Vela, A. Bux, B. Barbosa, and A. M. Herzallah, "Effects of technological, organizational, and environmental factors on social media adoption," *J. Asian Finance Econ. Bus.*, vol. 7, no. 10, pp. 989–998, 2020.
- [20] F. D. Davis, R. P. Bagozzi, and P. R. Warshaw, "User acceptance of computer technology: A comparison of two theoretical models," *Manage. Sci.*, vol. 35, no. 8, pp. 982–1003, 1989.
- [21] V. Venkatesh, M. G. Morris, G. B. Davis, and F. D. Davis, "User acceptance of information technology: Toward a unified view," *MIS Quart.*, vol. 27, no. 3, pp. 425–478, 2003.
- [22] I. Ajzen, "The theory of planned behavior," Organizational Behav. Hum. Decis. Processes, vol. 50, no. 2, pp. 179–211, 1991.
- [23] E. M. Rogers, *Diffusion of Innovations*, 4th ed. New York, NY, USA: Free Press, 1995.
- [24] M. Dacin, J. Goodstein, and W. Richard Scott, "Institutional theory and institutional change: Introduction to the special research forum," *Acad. Manage. J.*, vol. 45, no. 1, pp. 45–56, 2002.
- [25] C. L. Iacovou, I. Benbasat, and A. S. Dexter, "Electronic data interchange and small organizations: Adoption and impact of technology," *MIS Quart.*, vol. 19, pp. 465–485, 1995.
- [26] C. Oliver, "The influence of institutional and task environment relationships on organizational performance: The canadian construction industry," *J. Manage. Stud.*, vol. 34, no. 1, pp. 99–124, 1997.
- [27] J. Alzamora-Ruiz, M. Fuentes-Fuentes, and M. Martinez-Fiestas, "Together or separately? Direct and synergistic effects of effectuation and causation on innovation in technology-based SMEs," *Int. Entrepreneurship Manage. J.*, vol. 17, no. 4, pp. 1917–1943, 2021.
- [28] D. K. Nguyen, T. Broekhuizen, J. Q. Dong, and P. Verhoef, "Digital readiness: Construct development and empirical validation," in *Proc. Int. Conf. Inf. Syst.*, 2019, Art. no. 9.
- [29] J. Jöhnk, M. Weißert, and K. Wyrtki, "Ready or not, AI comes—An interview study of organizational AI readiness factors," *Bus. Inf. Syst. Eng.*, vol. 63, no. 1, pp. 5–20, 2021.
- [30] A. Caputo, G. Marzi, and M. M. Pellegrini, "The internet of things in manufacturing innovation processes: Development and application of a conceptual framework," *Bus. Process Manage. J.*, vol. 22, no. 2, pp. 383–402, 2016.
- [31] G. Adamson, L. Wang, M. Holm, and P. Moore, "Cloud manufacturing–a critical review of recent development and future trends," *Int. J. Comput. Integr. Manuf.*, vol. 30, no. 4-5, pp. 347–380, 2017.
- [32] M. Hänninen, A. Smedlund, and L. Mitronen, "Digitalization in retailing: Multi-sided platforms as drivers of industry transformation," *Baltic J. Manage.*, vol. 13, no. 2, pp. 152–168, 2018.
- [33] L. G. Tornatzky, M. Fleischer, and A. K. Chakrabarti, Processes of Technological Innovation. Lanham, MD, USA: Lexington Books, 1990.

- [34] H. O. Awa, O. U. Ojiabo, and L. E. Orokor, "Integrated technologyorganization-environment (TOE) taxonomies for technology adoption," *J. Enterprise Inf. Manage.*, vol. 30, no. 6, pp. 893–921, 2017.
- [35] S. Sun, D. J. Hall, and C. G. Cegielski, "Organizational intention to adopt big data in the B2B context: An integrated view," *Ind. Marketing Manage.*, no. 86, pp. 109–121, 2020.
- [36] S. Mittal, M. A. Khan, D. Romero, and T. Wuest, "A critical review of smart manufacturing & industry 4.0 maturity models: Implications for small and medium-sized enterprises (SMEs)," *J. Manuf. Syst.*, vol. 49, pp. 194–214, 2018.
- [37] C. Wang, Y. Fang, and C. Zhang, "Mechanism and countermeasures of 'the innovator's dilemma' in business model," *J. Innov. Knowl.*, vol. 7, no. 2, 2022, Art. no. 100169.
- [38] S. Winkelhaus and E. H. Grosse, "Logistics 4.0: A systematic review towards a new logistics system," *Int. J. Prod. Res.*, vol. 58, no. 1, pp. 18–43. 2020.
- [39] T. Beliaeva, M. Ferasso, S. Kraus, and E. J. Damke, "Dynamics of digital entrepreneurship and the innovation ecosystem: A multilevel perspective," *Int. J. Entrepreneurial Behav. Res.*, vol. 26, no. 2, pp. 266–284, 2020.
- [40] J. Holmström and J. Partanen, "Digital manufacturing-driven transformations of service supply chains for complex products," *Supply Chain Manage: An Int. J.*, vol. 19, no. no. 4, pp. 421–430, 2014.
- [41] M. J. Sousa and Á. Rocha, "Skills for disruptive digital business," J. Bus. Res., vol. no. 94, pp. 257–263, 2019.
- [42] M. S. Raisinghani and L. L. Meade, "Strategic decisions in supply-chain intelligence using knowledge management: An analytic-network-process framework," *Supply Chain Manage., Int. J.*, vol. 10, no. 2, pp. 114–121, 2005.
- [43] J. Cenamor, V. Parida, and J. Wincent, "How entrepreneurial SMEs compete through digital platforms: The roles of digital platform capability, network capability and ambidexterity," *J. Bus. Res.*, vol. 100, pp. 196–206, 2019.
- [44] N. Van Zeebroeck, T. Kretschmer, and J. Bughin, "Digital 'is' strategy: The role of digital technology adoption in strategy renewal," *IEEE Trans. Eng. Manage.*, early access, Jun., 11, 2021, doi: 10.1109/TEM.2021.3079347.
- [45] S. S. Kamble, A. Gunasekaran, and S. A. Gawankar, "Sustainable industry 4.0 framework: A systematic literature review identifying the current trends and future perspectives," *Process Saf. Environ. Protection*, vol. 117, pp. 408–425, 2018.
- [46] M. Ghobakhloo, "Determinants of information and digital technology implementation for smart manufacturing," *Int. J. Prod. Res.*, vol. 58, no. 8, pp. 2384–2405, 2020.
- [47] P. Mikalef and A. Pateli, "Information technology-enabled dynamic capabilities and their indirect effect on competitive performance: Findings from PLS-SEM and fsQCA," J. Bus. Res., vol. 70, pp. 1–16, 2017.
- [48] J. Karimi and Z. Walter, "The role of dynamic capabilities in responding to digital disruption: A factor-based study of the newspaper industry," J. Manage. Inf. Syst., vol. 32, no. 1, pp. 39–81, 2015.
- [49] D. J. Teece, "Business models and dynamic capabilities," Long Range Plan., vol. 51, no. 1, pp. 40–49, 2018.
- [50] K. Mahroof, "A human-centric perspective exploring the readiness towards smart warehousing: The case of a large retail distribution warehouse," *Int. J. Inf. Manage.*, vol. no. 45, pp. 176–190, 2019.
- [51] J. Baker, "The technology-organization-environment framework," in *In-formation Systems Theory*. Berlin, Germany: Springer, 2012, pp. 231–245.
- [52] T. Oliveira and M. F. Martins, "Literature review of information technology adoption models at firm level," *Electron. J. Inf. Syst. Eval.*, vol. 14, no. 1, pp. 110–121, 2011.
- [53] E. M. Rogers, *Diffusion of Innovations*, 5th ed. New York, NY, USA: Free Press, 2003.
- [54] T. Oliveira, M. Thomas, and M. Espadanal, "Assessing the determinants of cloud computing adoption: An analysis of the manufacturing and services sectors," *Inf. Manage.*, vol. 51, no. 5, pp. 497–510, 2014.
- [55] L. G. Tornatzky and K. J. Klein, "Innovation characteristics and innovation adoption-implementation: A meta-analysis of findings," *IEEE Trans. Eng. Manage.*, vol. EM-29, no. 1, pp. 28–45, Feb. 1982.
- [56] B. Ramdani, D. Chevers, and D. A. Williams, "SMEs' adoption of enterprise applications: A technology-organization-environment model," *J. Small Bus. Enterprise Develop.*, vol. 20, no. 4, pp. 735–753, 2013.
- [57] M. Ghobakhloo and N. T. Ching, "Adoption of digital technologies of smart manufacturing in SMEs," J. Ind. Inf. Integration, vol. 16, 2019, Art. no. 100107.
- [58] N. Tuptuk and S. Hailes, "Security of smart manufacturing systems," J. Manuf. Syst., vol. 47, pp. 93–106, 2018.

- [59] X. Neumeyer, S. C. Santos, and M. H. Morris, "Overcoming barriers to technology adoption when fostering entrepreneurship among the poor: The role of technology and digital literacy," *IEEE Trans. Eng. Manage.*, vol. 68, no. 6, pp. 1605–1618, Dec. 2021.
- [60] A. Molla and P. S. Licker, "E-Commerce adoption in developing countries: A model and instrument," *Inf. Manage.*, vol. 42, no. 6, pp. 877–899, 2005.
- [61] A. S. Bharadwaj, "A resource-based perspective on information technology capability and firm performance: An empirical investigation," *MIS Quart.*, vol. 24, no. 1, pp. 169–196, 2000.
- [62] L. Tang, Z. Gu, Q. Zhang, and J. Liu, "The effect of firm size, industry type and ownership structure on the relationship between firms' sustainable innovation capability and stock liquidity," *Operations Manage. Res.*, 2022.
- [63] D. A. Marchand, W. J. Kettinger, and J. D. Rollins, "Information orientation: People, technology and the bottom line," *MIT Sloan Manage. Rev.*, vol. 41, no. 4, 2000, Art. no. 69.
- [64] Y. M. Wang, Y. S. Wang, and Y. F. Yang, "Understanding the determinants of RFID adoption in the manufacturing industry," *Technol. Forecasting Social Change*, vol. 77, no. 5, pp. 803–815, 2010.
- [65] X. Duan, H. Deng, and B. Corbitt, "A critical analysis of e-market adoption in australian small and medium sized enterprises," in *Proc. Pacific Asia Conf. Inf. Syst.*, 2010, Art. no. 169.
- [66] S. Sharma and A. Rai, "An assessment of the relationship between ISD leadership characteristics and IS innovation adoption in organizations," *Inf. Manage.*, vol. 40, no. 5, pp. 391–401, 2003.
- [67] A. Parasuraman and C. L. Colby, "An updated and streamlined technology readiness index: TRI 2.0," J. Service Res., vol. 18, no. 1, pp. 59–74, 2015.
- [68] D. Hradecky, J. Kennell, W. Cai, and R. Davidson, "Organizational readiness to adopt artificial intelligence in the exhibition sector in western Europe," *Int. J. Inf. Manage.*, vol. 65, 2022, Art. no. 102497.
- [69] R. B. Bouncken, S. Kraus, and N. Roig-Tierno, "Knowledge- and innovation-based business models for future growth: Digitalized business models and portfolio considerations," *Rev. Managerial Sci.*, vol. 15, pp. 1–14, 2021.
- [70] S. Kraus, S. Durst, J. J. Ferreira, P. Veiga, P. N. Kailer, and A. Weinmann, "Digital transformation in business and management research: An overview of the current status quo," *Int. J. Inf. Manage.*, vol. 63, 2022, Art. no. 102466.



Nessrine Omrani received the Ph.D. degree in economics from the University of Paris-Saclay, Gif-sur-Yvette, France, in 2012, and the Postdoctoral degree (HDR) in management from the University of Paris-Est, Champs-sur-Marne, France, in 2020.

She is currently a Full Professor of digital transformation and the Director of Management Department, Paris School of Business, Paris, France. She has authored or coauthored in various academic journals, such as *Journal of Economic Literature*, *Technological Forecasting and Social Change*, IEEE TRANS-

ACTIONS ON ENGINEERING MANAGEMENT, Information Economics and Policy, European Journal of Comparative Economics, and Economic and Industrial Democracy. Her research interests include the areas of digital transformation, online consumer behavior, crowdfunding, and privacy.



Nada Rejeb received the Ph.D. degree in management science (strategic management) from the University of Sfax, Sfax, Tunisia, in 2011.

She is currently an Associate Professor in entrepreneurship and management at ICD Business School of Paris, Paris, France. She is Visiting Professor with the JAMK School of Business, Jyväskylä, Finland, and University of Nanterre, Paris, France. Her research interests include knowledge networks, innovation, and collaborative relationships, with a specific interest in family firms' collaboration strate-

gies. She is also interested in topical issues in entrepreneurship, typically, health in entrepreneurship and entrepreneurial entry by disadvantaged minorities. Her works have been published in leading journals in innovation, entrepreneurship, and small business, such as *Small Business Economics* and *Journal of Business Research*.



Adnane Maalaoui received the Ph.D. degree in management from IAE Toulon, Toulon, France, in 2009. He is currently a Professor in entrepreneurship and the Head of the entrepreneurship programs with IPAG Business School, Paris, France. His research intersts include issues within the entrepreneurship

field, and in particular, exploring disadvantaged entrepreneurs (such as elderly, immigrant, and disabled entrepreneurs). He also has expertise within the areas of entrepreneurial intention and the cognitive approach to entrepreneurship. He mainly applies those

questions to cases of diversity. He is the author of more than 20 articles published in academic journals, has authored or coauthored articles published in professional journals and in edited books, and is the author of a series of French language MOOCs on entrepreneurship.



Sascha Kraus received the doctoral degree in social and economic sciences from Klagenfurt University, Klagenfurt, Austria, in 2006, the Ph.D. degree in industrial engineering and management from Helsinki University of Technology, Espoo, Finland, in 2009, and the Habilitation (Venia Docendi) in Management from Lappeenranta University of Technology, Lappeenranta, Finland, in 2010.

He is currently Full Professor of Management with the Free University of Bozen-Bolzano, Bolzano, Italy, and Distinguished Visiting Professor (SARChI En-

trepreneurship Education) at the University of Johannesburg, Johannesburg, South Africa. Before that, he was a Full Professor with Utrecht University, Utrecht, The Netherlands, the University of Liechtenstein, École Supérieure du Commerce Extérieur, Paris, France, and Durham University, Durham, U.K. He was also a Visiting Professor with Copenhagen Business School, Frederiksberg, Denmark, and the University of St. Gallen, Gallen, Switzerland.



Marina Dabić received the M.Sc. degree in economics, the M.Sc. degree in marketing, and Ph.D. degree in economics from the University of Zagreb, Zagreb, Croatia, in 1983, 1990, and 2000, respectively.

She is currently a Full Professor of entrepreneurship and international business with the Faculty of Economics and Business, University of Zagreb, the School of Business and Economics, University of Ljubljana, Ljubljana, Slovenia, and University of Dubrovnik, Dubrovnik, Croatia. She has authored or

coauthored articles published in a wide variety of international journals, including the Journal of International Business Studies, Journal of World Business, Journal of Business Research, Technological Forecasting and Social Change, Technovation, Small Business Economics, Small Business Management Journal, International Journal of Human Resource Management, IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT, Journal of Business Ethics, among many others.

Prof. Dabić is an Associate Editor for the *Technological Forecasting and Social Change*, Departmental Editor of IEEE TRANSACTIONS ON ENGINEERING MANAGEMENT, and an Associate Editor for the *Technology in Society*. She was a Member at large for the IEEE-TEMS 2020-2022. She has been a grant holder of TEMPUS FoSentHE and participated as a partner in more than 15 projects granted by the European Commission. She is the WP Leader for Industry 4.0 in the Horizon 2020 RISE Open innovation project, Interreg Wool project, and ERASMUS K2 VOIS project.

Open Access provided by 'Libera Università di Bolzano' within the CRUI CARE Agreement